Author's response to reviews

Title: In vivo activity of terpinen-4-ol, the main bioactive component of Melaleuca alternifolia Cheel (tea tree) oil against azole-susceptible and -resistant human pathogenic Candida species

Authors:
Francesca Mondello (mondello@iss.it)
Flavia De Bernardis (fdberna@iss.it)
Antonietta Girolamo (girolamo@iss.it)
Antonio Cassone (cassone@iss.it)
Giuseppe Salvatore (pino@iss.it)

Version: 5 Date: 3 November 2006

Author's response to reviews: see over
MANUSCRIPT 1080985485109619 - In vivo activity of terpinen-4-ol, the main bioactive component of Melaleuca alternifolia Cheel (tea tree) oil against azole-susceptible and -resistant human pathogenic Candida species.

Francesca Mondello, Flavia De Bernardis, Antonietta Girolamo, Antonio Cassone and Giuseppe Salvatore

Dear Editor

We wish to inform you that we have made further formatting changes to the final document and we are now satisfied that the document is ready for publishing.

We wish to thank you for your kind consideration.

Kind regards

Dr. Francesca Mondello

Department of Infectious, Parasitic and Immune-mediated Diseases
Istituto Superiore di Sanità
(National Health Institute)
Viale Regina Elena, 299
00161 Roma (Italy)
Tel ++39-06-49902654
Fax++39-06-49387112
e-mail: mondello@iss.it